Author’s response to reviews

Title: A systematic comparison of copy number alterations in four types of female cancer

Authors:

Fatemeh Kaveh (fatemeh.kaveh@medisin.uio.no)
Lars Baumbusch (lars.o.baumbusch@rr-research.no)
Daniel Nebdal (Daniel.Nebdal@rr-research.no)
Anne-Lise Børresen-Dale (a.l.borresen-dale@medisin.uio.no)
Ole Lingjærde (ole@ifi.uio.no)
Hege Edvardsen (hege.edvardsen@rr-research.no)
Vessela N.Kristensen (v.n.kristensen@medisin.uio.no)
Hiroko K.Solvang (hiroko.solvang@imr.no)

Version: 1 Date: 27 Sep 2016

Author’s response to reviews:

Dear Dr. Rice,

Please find enclosed the revised version of our manuscript.

We have carefully address all editorial requests:

1. We have removed the ages from supplementary table 1 (and all other places) and replaced it by an age range.

2. We have up-loaded a clean version of the manuscript without tracked changes or highlighting.

3. We would like to notify that we have revised the manuscript carefully in order to remove all spelling mistakes, repetitive words, confusing phrases or missing links. The "language clean-up" has not changed any results or any conclusions.
We would like to express our gratitude for all your understanding, support, and help during this submission process. This is highly appreciated.

Best regards

Fatima and Lars